Author's response to reviews

Title: Coronary artery surgery in a patient with achondroplasia- a rare case report

Authors:

Georgios I Tagarakis (htagarakis@hotmail.com)
Andony J Baddour (badour@otenet.gr)
Nicholas Desimonas (ndesimonas@gmail.com)
Dimos Karangelis (dimoskaragel@yahoo.gr)
Angheliki Tsantsaridou (tsantsar@hotmail.com)
Ioannis Tentzeris (y_tentzeris@yahoo.com)
Nikolaos B Tsilimingas (tsilimingas@med.uth.gr)

Version: 3 Date: 2 May 2010

Author's response to reviews:

Dear editor,

First of all we wish to thank you for giving us the opportunity for revision of our present manuscript. Indeed the reviewers’ remarks are significantly important for the paper’s improvement.

In accordance with the constructive criticism and very helpful pieces of advice from the reviewers we prepared a substantial revision. In the revised manuscript we made every effort to meet the criteria raised by the reviewers’ comments. We responded to all reviewer comments point by point (see below). Changes in the manuscript are marked in red. We hope that the revised manuscript now meets the demands for publication in your Journal.

Kind regards,

Tagarakis George MD

Response to Reviewer 1:
1. We have added details on the patient’s history.
2. We have provided the patient’s lipid-biochemical markers’ profile although we had already emphasized on the patient’s lack of predisposing factors.
3. We have given details about echocardiography and angiography findings.

Response to Reviewer 2: We thank the reviewer for his comments.